

JGR Oceans

RESEARCH ARTICLE

10.1029/2020JC016114

Key Points:

- Our model experiments support the established view that deep water formation does not play a major role in Baltic Sea bottom water renewal
- We implement a subgrid-scale brine rejection parameterization; its effect is too weak to impact deep water exchange in the northern Baltic
- Hypothesized bottom water ventilation through convection during cold climate phases cannot be explained by reduced air temperatures only

Supporting Information:

Supporting Information S1

Correspondence to:

C. Giesse, celine.giesse@uni-hamburg.de

Citation:

Giesse, C., Meier, H. E. M., Neumann, T., & Moros, M. (2020). Revisiting the role of convective deep water formation in northern Baltic Sea bottom water renewal. *Journal of Geophysical Research: Oceans, 125,* e2020JC016114. https://doi.org/ 10.1029/2020JC016114

Received 25 JAN 2020 Accepted 21 SEP 2020 Accepted article online 22 SEP 2020

[Correction added on 15 OCT, after first online publication: Projekt Deal funding statement has been added.]

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Revisiting the Role of Convective Deep Water Formation in Northern Baltic Sea Bottom Water Renewal

Céline Giesse^{1,2,3}, H. E. Markus Meier^{1,4}, Thomas Neumann¹, and Matthias Moros¹

¹Leibniz Institute for Baltic Sea Research Warnemünde (IOW), Rostock, Germany, ²Institute of Oceanography, CEN, Universität Hamburg, Hamburg, Germany, ³International Max Planck Research School on Earth System Modelling, Max Planck Institute for Meteorology, Hamburg, Germany, ⁴Swedish Meteorological and Hydrological Institute, Norrköping, Sweden

Abstract Deep water renewal and ventilation of the Baltic Sea is commonly considered to occur solely via saltwater inflows from the North Sea. However, recent analysis of geophysical and sediment proxy data suggests convective deep water formation during wintertime as a second process that has contributed to the ventilation of the northern and central Baltic Sea bottom waters during cold climate periods. Here, we investigate the role of deep water formation in the northern Baltic Sea in a regional ocean circulation model. Selecting the particularly cold winter 1986/1987 as a reference period, we perform sensitivity experiments in which the atmospheric temperature forcing is changed and/or localized brine rejection on subgrid scales is accounted for through a parameterization. We study the sinking and circulation of water masses via passive tracers introduced into the model. Generally, our model results support the established view that convective deep water formation does not play a major role in Baltic Sea deep water renewal. While a reduction of air temperatures does not qualitatively change the sinking and circulation of water masses in our experiments, brine rejection could potentially lead to localized deep water formation. However, the impact is too weak to possibly change the large-scale deep water exchange between the Baltic proper and the northern Baltic sub-basins. Although being in line with established knowledge, through consideration of the model limitations, our results provide insights on how and under which circumstances convective deep water formation could potentially have occurred during cold climate periods.

Plain Language Summary The transport of oxygen-rich water into the deep parts of the Baltic Sea is of crucial importance for the ecosystem, as under low-oxygen conditions (hypoxia), most organisms cannot survive and so-called dead zones are formed. The ventilation of Baltic Sea bottom waters is commonly considered to occur solely through inflows of dense, oxygenated water from the North Sea. However, recent analysis of sediment cores and geophysical data suggests the contribution of a second process during cold climate periods: convective deep water formation. This process is initiated through surface cooling or an increase of surface water salinity during sea ice formation (brine rejection) in the winter season. Here, we investigate the role of deep water formation in the northern Baltic Sea in a regional ocean circulation model. We test the influence of reduced air temperatures and consideration of strong, localized brine rejection by performing sensitivity experiments. Generally, our model results do not support the hypothesis of convective deep water formation contributing to the renewal of Baltic Sea bottom waters, but are in line with the established view. Nevertheless, our results provide insights on how and under which circumstances convective deep water formation could potentially have occurred during cold climate periods.

1. Introduction

The Baltic Sea (Figure 1) is a semi-enclosed, brackish sea which is connected to the North Sea via a shallow and narrow connection, the Danish straits. Limited water exchange with the North Sea and a positive freshwater budget from river discharge and net precipitation result in strong horizontal and vertical salinity gradients and the establishment of an estuarine-like circulation with outflowing brackish water at the surface and inflowing saltier water close to the seafloor. The permanent and pronounced halocline constrains an exchange between surface and bottom waters and, as a consequence, facilitates frequent occurrence of suboxic or anoxic conditions in the bottom waters. Ventilation and renewal of the deep waters of the central Baltic Sea is commonly considered to occur solely through inflows of dense and oxygenated water from the





Figure 1. Map and bathymetry of the Baltic Sea. Sub-basin names relevant for this study are indicated.

North Sea (e.g., Meier et al., 2006). These inflows occur episodically in response to specific meteorological conditions (Schinke & Matthäus, 1998). An effective oxygenation of the central Baltic Sea can occur through strong inflow events, so-called major Baltic inflows (Fischer & Matthäus, 1996; Matthäus & Franck, 1992) or sequences of smaller inflows (Neumann et al., 2017). The northern central Baltic Sea is connected to two major sub-basins, namely, the Gulf of Bothnia in the north, which is further divided into the Bothnian Sea and the Bothnian Bay, and the Gulf of Finland in the northeast. The water exchange between the Baltic proper and these sub-basins is commonly considered to be of an estuarine type as well with inflows of surface water from the central Baltic Sea leading to a deep water renewal in the northern sub-basins (e.g., Andrejev et al., 2004; Elken et al., 2003; Marmefelt & Omstedt, 1993; Myrberg & Andrejev, 2006).

In addition to this well-established view, recent findings by Moros et al. (2020) suggest wintertime deep water formation as a second process that has contributed to sediment re-deposition in and bottom water ventilation of the northern and central Baltic Sea during preceding cold phases like the Little Ice Age (LIA). This hypothesis is based on evidence from numerous sediment cores (e.g., irregularities in chronologies based on radiocarbon dating results) and seismoacoustic profiles (examples are shown in the supporting information), indicating that during cold phases, local convection resulted in deep water formation and further produced

strong southward-directed bottom currents causing a lateral sediment influx from the northern sub-basins to the northern Baltic proper and Landsort Deep (Moros et al., 2020). Furthermore, measurements of oxygen concentration indicate that bottom water ventilation through wintertime deep water formation might have also occurred during severe winters in the current climate state (see supporting information).

The hypothesis that convection is a key process for the deep water renewal of the northern Baltic Sea is not new and has already been raised by Granqvist (1938). Deep water formation occurs as a consequence of the creation of dense water at the ocean surface, being less buoyant than the water below, and subsequent convection. The necessary density increase or buoyancy loss can be caused either by temperature reduction (i.e., thermal convection) or salinity increase (i.e., haline convection). However, the freezing temperature in the northern Baltic is below the temperature of maximum density (TMD) due to the low salinity. That is, cooling below the TMD will increase buoyancy and temperature sensitivity of buoyancy is low close to the TMD. Therefore, although the salinity in the northern Baltic Sea is low, haline convection should be the major process controlling wintertime convection. A substantial increase of surface salinity can be generated during the sea ice period through the process of brine rejection. During the formation of sea ice, the salt gets rejected from the forming ice crystals and accumulates as highly saline water, called brine. The brine gets trapped within the interstices of the sea ice and is gradually released into the surface water through channels and holes within the ice via different mechanisms (e.g., Notz & Worster, 2009; Untersteiner, 1968). Marmefelt and Omstedt (1993) investigated the potential of thermal and haline convection for deep water renewal in the Gulf of Bothnia based on mean vertical salinity and temperature profiles from observational data. They concluded that neither deep thermal nor deep haline convection is likely to occur frequently and that deep water renewal occurs predominantly through inflows from the Baltic proper. However, they also noted distinct occasions at which deep, bottom-reaching convection could potentially have occurred, namely, in 1979 for thermal convection and in the winter 1986/1987 for haline convection, and pointed out the need for further wintertime observations. Recent ship-based, snapshot observations of sea ice, brine, and water column profiles in the ice-covered Bothnian Bay in March 2017 indicate a ventilation through lateral inflows rather than convection (Neumann et al., 2020), supporting the conclusion of Marmefelt and Omstedt (1993).

The conclusion drawn by Marmefelt and Omstedt (1993) that deep haline convection is unlikely to occur in the Gulf of Bothnia is based on simple budget considerations, assuming that all rejected salt is homogeneously mixed within the surface layer. This approach neglects the heterogeneity of the sea ice cover and

the possibility of strong localized convection. Areas with particularly high brine-induced buoyancy fluxes are openings in the sea ice cover, like leads and polynyas, as their exposure to the cold atmosphere allows for fast ice growth. Although making up only a small fraction of the total ice cover, leads account for a major fraction of the salt input into the ocean associated with brine rejection (e.g., Maykut, 1978). The convection process in leads has been extensively studied by means of observations (e.g., Morison & McPhee, 1998; Morison et al., 1992; Muench et al., 1995) and numerical modeling (e.g., Kozo, 1983; Smith & Morison, 1993, 1998; Smith et al., 2002). It is associated with the formation of convective plumes which, in case of little ice motion, sink to the bottom of the mixed layer, only partially mixing with ambient water, and then spread horizontally away from the lead center. In case of low stratification, the plumes might even penetrate through the pycnocline and sink to the bottom of the sea. Taking this localized brine rejection into account, it seems worthwhile to reconsider the results of Marmefelt and Omstedt (1993).

In this study, the role of deep water formation in the northern Baltic Sea is revisited in a model approach using a regional configuration of an ocean circulation model. The guiding question is whether a ventilation of bottom waters in the northern sub-basins and the northern central Baltic Sea due to wintertime deep water formation is present in the model. It is further subdivided into the following research questions, each addressed by different model experiments.

- 1. Is wintertime deep water formation occurring in the northern Baltic Sea sub-basins?
- 2. What is the possible effect of strong, localized brine rejection in the Baltic Sea?
- 3. How does the situation change in a colder climate state?

In order to represent localized brine rejection and convection on scales smaller than the model grid size, a subgrid-scale parameterization of brine rejection (Nguyen et al., 2009) is implemented into the ocean model and applied to the Baltic Sea for the first time.

2. Methods

2.1. Model Description

The numerical model used in this study is based on the circulation model Modular Ocean Model (MOM) 5.1 (Griffies, 2004) and has been adapted to the Baltic Sea with an open boundary condition to the North Sea. Vertical subgrid mixing is calculated with the non-local K-profile scheme (Large et al., 1994). Horizontal viscosity is calculated from the Smagorinsky (1963) scheme. The MOM model is complemented with a sea ice model that accounts for the formation of sea ice and the ice drift (Winton, 2000). Brine rejection is parameterized by a freshwater flux during ice formation considering a bulk ice salinity. The horizontal resolution of the model grid is one nautical mile, while vertically, the model is resolved into 152 layers, starting with 0.5-m layer thickness at the surface and gradually increasing with depth up to a layer thickness of 2 m. The model was integrated from 1950 until 2018. Meteorological forcing was derived from the coastDat2 data set (Geyer, 2014) and the runoff is based on HELCOM (www.helcom.fi) data. For this study, we select the years 1986 and 1987, representing a particularly cold winter with the largest ice extent in the current climate period (see supporting information), and perform a sensitivity study (see section 2.4).

2.2. Subgrid-Scale Brine Rejection Parameterization

Due to openings and different ice thicknesses, the surface boundary conditions in the presence of a sea ice cover are very heterogeneous. In particular, brine rejection in leads causes strong buoyancy losses on scales of only a few to a few hundred meters, that are generally not resolved in ocean circulation models. Without any special treatment, the surface buoyancy flux is averaged horizontally over an entire grid cell, suppressing the potential of localized deep mixing. The consequences are too shallow penetration depths of the convective plumes and too strong mixing and homogenization above the penetration depth, as shown for instance by Losch et al. (2006) and Jin et al. (2012).

In order to represent the subgrid-scale convection associated with brine rejection in leads, the simple concept of brine rejection parameterizations (BRPs) has been introduced (Duffy & Caldeira, 1997; Duffy et al., 1999; Nguyen et al., 2009 and follow-up studies). The basic idea is to distribute the rejected salt vertically within the mixed layer, mimicking the spreading through a convective plume, instead of placing it in the top layer and treating it via a vertical mixing scheme. This is of course an oversimplified representation of subgrid-scale convection, neglecting for instance the mixing of heat and other tracers, but it provides a first,





simple approach to overcome the problem of horizontally averaged buoyancy fluxes. Here, the formulation of Nguyen et al. (2009), which is based on knowledge of the convection process in leads from observations as well as laboratory and numerical experiments, is followed. The rejected salt is distributed vertically according to a power law distribution, that is,

$$s(z) = \begin{cases} Az^n & \text{if } z \le D_{\text{sp}} \\ 0 & \text{if } z > D_{\text{sp}} \end{cases},$$
(1)

where *z* is the depth, defined as being positive below sea level for simplicity reasons, s(z) is the salt distribution function, *n* is the distribution power, and D_{sp} is the salt plume depth. The constant *A* is determined by the total amount of rejected salt S_0 :

$$\int_{0}^{D_{\rm sp}} s(z) dz = \int_{0}^{D_{\rm sp}} A z^{n} dz = \frac{A}{n+1} D_{\rm sp}^{n+1} \stackrel{!}{=} S_{0}$$
(2)

$$\Rightarrow A = \frac{n+1}{D_{\rm sp}^{n+1}} S_0. \tag{3}$$

Figure 2. Mask showing the release areas of the three passive tracers, marking waters of the three northern sub-basins: the Bothnian Bay (BoB, red), the Bothnian Sea (BoS, blue), and the Gulf of Finland (GoF, green). Further shown are the stations BO3, SR5, LL7, and BY31 (marked as yellow dots) as well as three zonal transects within the sub-basins (black lines), used for evaluation purpose.

The distribution power *n* and the salt plume depth D_{sp} are adjustable parameters. For their specific application area, the Arctic Ocean, Nguyen et al. (2009) found that setting n = 5 and D_{sp} to a depth corresponding to a density gradient of $d\rho/dz = 0.002 \text{ kg/}m^4$ yields the lowest model-data misfits. In this study, the value of n = 5 for the distribution power is adopted from Nguyen et al. (2009) and two different methods for determining the salt plume depth D_{sp} are used (see section 2.4).

Note that, apart from the concept of BRPs, other approaches for taking into account subgrid-scale convection exist, for instance two- or multi-column mixing schemes (e.g., Barthélemy et al., 2016; Jin et al., 2015).

2.3. Passive Tracers

One main aim of this study is to track the pathways of water masses originating in the northern Baltic Sea sub-basins in order to verify if wintertime deep water formation is happening and whether these water masses are entering the central Baltic Sea.

Therefore, three passive tracers, marking water of the Bothnian Bay (BoB), the Bothnian Sea (BoS, including also the Åland and Archipelago Seas), and the Gulf of Finland (GoF) as shown in Figure 2, are introduced into the model. Depending on the addressed research question of the respective experiment, two different kinds of passive tracers, which differ in the method of initialization, are used in this study.

2.3.1. Surface Tracer

In order to answer the first research question, whether wintertime deep water formation is occurring in the northern Baltic Sea sub-basins, the passive tracers are initialized as so-called "surface tracers." Their concentration is set to $1 m^{-3}$ in the surface cells within the initialization area (shown in Figure 2) at each time step during the period of November to March. Afterwards, they are advected freely. This enables to qualitatively follow the pathways of the winter surface water masses of the different basins and to observe possible deep water formation through either haline or thermal convection.

2.3.2. Brine Tracer

In order to answer the second research question, which is explicitly targeting the potential of localized brine rejection for deep water renewal in the northern Baltic Sea, the passive tracers are initialized as so-called "brine tracers." Their concentration is set to the actual salinity caused by salt rejected during ice formation. The tracer concentration has the unit g/kg and is a quantitative measure of brine rejection. The advantage of this method is that the subgrid-scale BRP, introduced in section 2.2, can be applied not only to the salinity itself but also to the brine tracer concentration. This allows us to follow the pathways of water masses at



Table 1	L
---------	---

List of Sensitivity Experiments

Experiment	Atmospheric forcing	Brine rejection parameterization
ref BRP bottom BRP MLD Cold Climate BRP bottom Cold Climate BRP MLD Cold Climate	Winter 1986/1987 Winter 1986/1987 Winter 1986/1987 –2 K air temperature –2 K air temperature	No BRP $D_{sp} = bottom$ $D_{sp} = mixed layer depth$ No BRP $D_{sp} = bottom$ $D_{sp} = mixed layer depth$

depths corresponding to the sinking depths of the convective brine plumes, assuming that they behave as prescribed by the BRP.

2.4. Sensitivity Experiments

In total, six experiments are conducted. An overview of the experiments is given in Table 1.

The first experiment is a reference experiment ("ref") with the atmospheric forcing of the years 1986/1987 and without a BRP. On the one hand, it aims to answer the first research question, namely, whether there is wintertime deep water formation (either through thermal or haline convection) occurring in the northern Baltic Sea. On the other hand, it serves as a reference experiment for the following sensitivity experiments.

The next two experiments address the second research question, aiming to analyze the effect of strong, localized brine rejection. Therefore, the BRP (section 2.2) is switched on. The value of n = 5 for the distribution power is adopted from Nguyen et al. (2009). Two experiments with different salt plume depths D_{sp} , namely, the bottom ("BRP bottom") and a variable mixed layer depth ("BRP MLD"), are performed. The mixed layer depth is determined in the MOM model as the depth at which a critical buoyancy difference of 0.0003 m/s² with respect to the surface is reached.

In order to answer the third research question of how the situation is changed in a colder climate state, the three experiments are repeated with a different atmospheric forcing, in which the 2-m air temperature is reduced by $\Delta T = 2$ K constantly in space and time ("Cold Climate" experiments). With the change in air temperature, the specific humidity needs to be adjusted as well. Assuming that the relative humidity is unchanged, the specific humidity decreases with temperature according to the Clausius-Clapeyron relation, that is,

$$s_2 = s_1 \cdot \exp\left(\frac{\Delta H_{\text{vap}}}{R} \left(\frac{1}{T_1} - \frac{1}{T_2}\right)\right),\tag{4}$$

where s_1 and T_1 are the original specific humidity and temperature, s_2 is the new specific humidity corresponding to the new temperature T_2 , ΔH_{vap} is the enthalpy/heat of vaporization of water, and *R* is the universal gas constant. The temperature reduction by 2 K corresponds to the mean sea surface temperature (SST) difference between the LIA period and the modern warm period found by Kabel et al. (2012) based on a reconstruction from sediment proxy data. In that sense, it can be regarded as a change to LIA temperature forcing, assuming that SST and air temperature behave similarly. However, this might not be the case in the presence of an ice cover and severe winters during the LIA could have been much colder. Furthermore, in the Cold Climate experiments, we only account for differences in the temperature forcing, but not for other potential differences between different climate states, such as wind forcing or river runoff impacting the hydrographic conditions.

The "brine tracers" are introduced in all of the experiments, while the "surface tracers" are only used in the two experiments without BRP ("ref" and "Cold Climate").

3. Results

3.1. Model Validation

A reasonable model performance is the basis for confidence in the obtained model results. In the supporting information, we show a detailed validation of the simulation results by comparing sea ice, temperature, and salinity data with observations. Overall, the observations are well reproduced by the model, particularly for sea ice and temperature, but salinity biases lead to an overestimation of the stratification in the central Baltic Sea and Bothnian Sea.

3.2. Wintertime Deep Water Formation in the Northern Basins 3.2.1. Convective Sinking

The first question to answer is whether wintertime deep water formation through convection is occurring in the northern Baltic Sea sub-basins within the chosen model framework during the simulated winter





Figure 3. Winter (JFM) mean surface buoyancy flux in the reference experiment and its different contributions. (a) Total buoyancy flux, (b) thermal contribution, and (c) haline contribution from brine rejection.

1986/1987. Convection is generally forced by a loss of buoyancy at the water surface, which can either be temperature-driven (thermal convection) or salinity-driven (haline convection). A measure for the forcing of convection is the surface buoyancy flux, which can be expressed in terms of heat and freshwater fluxes as

$$B_0 = \frac{g\alpha_T}{\rho_0 c_p} Q_{\rm H} + g\beta_S S Q_{\rm FW},\tag{5}$$

where g is the gravitational acceleration, ρ_0 is a reference density, c_p is the heat capacity of seawater, S is the surface salinity, α_T and β_S are the thermal and haline expansion coefficients, and Q_H and Q_{FW} are the net heat and freshwater fluxes. Defining the fluxes as being directed from the atmosphere to the ocean, a positive buoyancy flux stabilizes the water column, whereas a negative buoyancy flux gives rise to convection. Figure 3 shows (a) the winter mean surface buoyancy flux, (b) the thermal contribution, and (c) the haline contribution from brine rejection (which is about one order of magnitude larger than the contribution from precipitation and evaporation) in the reference experiment. Here and in the following, "winter mean" denotes the average over the months from January to March (JFM), which are the months of major ice formation. Due to the TMD being higher than the freezing temperature in the brackish Baltic Sea water, the thermal expansion coefficient is negative when the SST sinks below the TMD and, despite the negative heat flux, the average thermal contribution in winter is positive over the whole northern Baltic Sea. The negative, haline contribution from brine rejection is, however, dominant in the coastal areas of the northern Baltic Sea, where most of the sea ice formation takes place, and therefore, the total buoyancy flux is negative in large parts of the northern Baltic Sea. The forcing would therefore allow for potential wintertime deep water through haline convection in coastal areas of the northern Baltic Sea.

Apart from the surface buoyancy forcing, the stratification conditions in the water column determine whether deep-reaching convection can occur. The introduction of the passive surface tracers allows for an analysis of the sinking of winter surface water in the individual sub-basins. Figure 4 shows a map of the sinking depth, defined as the depth at which the winter mean surface tracer concentration has decreased to 0.1 m^{-3} , that is, 10% of the initial concentration at the surface, in the reference experiment. In most parts of the northern Baltic Sea, the sinking depth does not exceed the model halocline depth of around 50–60 m. Exceptions are the Åland Sea and the Bothnian Bay, where sinking depths of locally up to 140 and 110 m, respectively, are found. While for the Bothnian Bay, this means that the tracer is reaching the bottom of the basin, the largest depths of the Åland Sea (up to 300 m) are not reached by convection. Note that here the sum of all three passive tracers is used to determine the sinking depth and tracer inflowing from other areas can increase the sinking depth from pure vertical convection, for example within the inflow channel from the Åland Sea into the Bothnian Sea.

A more detailed view of the convective sinking in the individual basins can be gained from analyzing vertical profiles of surface tracer concentration. The solid lines in Figure 5 show vertical profiles of winter mean





Figure 4. Sinking depth, defined as the depth at which the winter (JFM) mean surface tracer concentration has decreased to $0.1 m^{-3}$, in the reference experiment.

surface tracer concentration at stations in the Bothnian Bay (F9 and BO3), Bothnian Sea (SR5 and US3), and Gulf of Finland (LL7) in the reference experiment. Furthermore, for each of the basins, a vertical cut of winter mean surface tracer concentration is shown along a zonal transect through the basin (station and transect locations are marked in Figures 2 and 4). As can be seen from these exemplary profiles, the sinking behavior of surface tracer is different in each of the basins. In the Gulf of Finland, which is the most shallow of the three sub-basins (average depth of 38 m), the surface tracer reaches very close to the bottom. The concentration gradually decreases below a depth of around 40 m, but as the bottom in most parts of the basin is not much deeper as this mixed layer depth, measurable amounts of surface tracer can be found throughout the whole water column. In the Bothnian Sea, the surface tracer is not sinking below the halocline depth of around 50 m. The concentration gradient in the halocline layer varies at different locations within the Bothnian Sea, but nowhere the convection is sufficiently strong to bring surface water down to the bottom of the basin. The narrow channel with higher sinking depths, visible in Figure 4, is a result of tracer transport from the Åland Sea and not of local convection.

In contrast to that, significant amounts of surface tracer concentration reach the basin bottom in parts of the Bothnian Bay. As can be seen from the sinking depths in Figure 4 and the profiles in Figure 5, the primarily affected region is in the more southern part of the Bothnian Bay around station BO3. The BO3 tracer profile indicates that the deep water tracer

input is not a result of local convection, which would go along with a homogeneous water column, but rather the result of a lateral intrusion or inflow. The tracer concentrations along the zonal transect through the Bothnian Bay (Figure 5) reveal that the sinking of tracer occurs along the eastern coast. Two processes could potentially be responsible for the creation of a dense water mass inducing the convective sinking: mixing of Bothnian Bay surface water with (i) inflowing, more saline Bothnian Sea surface water or (ii) brine released during the formation of sea ice along the shallow coastal area. It is not possible to disentangle these two processes only from BoB tracer concentrations. However, we also find surface tracer from the Bothnian Sea at similar depths as the surface tracer from the Bothnian Bay and find qualitatively similar tracer sinking in an additional experiment where the process of brine rejection is turned off by setting the ice-ocean freshwater flux to zero (not shown). Therefore, we conclude that the dominant process for the simulated deep water formation in the Bothnian Bay is advection from the Bothnian Sea. Nevertheless, brine rejection might enhance the convective sinking.

3.2.2. Tracer Inflow Into the Central Baltic Sea

The geological findings described in Moros et al. (2020) suggest that during cold periods, not only local wintertime convection is taking place, but the deep water formed in the northern sub-basins is further transported into the deep parts of the northern central Baltic Sea. By evaluating the temporal evolution of tracer concentrations, we can follow the pathways of the winter surface water of the three sub-basins. This is shown in Figure 6 as snapshots of tracer concentrations at the bottom of the Baltic Sea in the reference experiment. In agreement to other model studies and observations (e.g., Andrejev et al., 2004; Myrberg & Andrejev, 2006), all sub-basins show a mean cyclonic (i.e., anti-clockwise) circulation. The BoB tracer leaves the Bothnian Bay southward along the western coast in the surface layer. The BoS tracer exits the Bothnian Sea partly northward, entering the Bothnian Bay bottom layer along the eastern coast, and partly southward through the Åland Sea into the surface layer of the Baltic proper. The GoF tracer escapes the Gulf of Finland at the northern part of its connection to the Baltic proper, partly entering the Bothnian Sea through the Archipelago Sea and partly flowing into the central Baltic Sea.

To evaluate the inflow of winter surface water from the northern sub-basins into the Baltic proper, we choose to analyze the tracer concentrations at Landsort Deep (station BY31). Figure 7 shows the time series of 30-day running mean tracer concentration of BoS and GoF surface tracer at station BY31. The time series of BoB tracer is not shown as the simulation period is too short for significant amounts of BoB tracer to arrive





Figure 5. Winter (JFM) mean vertical profiles of the three passive surface tracers at selected stations within the respective basin of release (from left to right: Bothnian Bay, Bothnian Sea, and Gulf of Finland) in the reference experiment (solid lines) and the Cold Climate experiment (dashed lines). The insets show winter mean surface tracer concentrations in a transect across each of the basins for the reference experiment.

at Landsort Deep. The travel distances from the Bothnian Sea and the Gulf of Finland are shorter. The tracers from these basins arrive at Landsort Deep from January on, but the highest concentrations are found in the summer months from June on. Both tracers are restricted to the surface layer above the halocline at around 60 m. BoS tracer concentrations are higher than GoF tracer concentrations. In winter and spring, the tracer concentrations are vertically relatively homogeneous due to a well-mixed surface layer; in summer, the surface layer is more stratified. Concluding, there is no inflow of deep water formed during winter in the northern sub-basins into the deep parts of the central Baltic Sea in the reference experiment.

3.3. Effect of Localized Brine Rejection

In the reference experiment, bottom-reaching deep water formation is occurring only in parts of the Bothnian Bay. It might, however, be possible that localized brine rejection on scales that are not resolved by the model (e.g., in sea ice leads) actually leads to stronger deep water formation. In order to test the potential effect of such localized brine rejection, we conducted two sensitivity experiments with a subgrid-scale BRP. In the "BRP bottom" experiment, all salt rejected during ice formation is mixed down to the bottom, which sets an upper limit for the potential effect of localized brine rejection. In the "BRP MLD" experiment, the rejected salt is mixed down to some mixed layer depth as it is done for instance in Nguyen et al. (2009) for the Arctic Ocean. Although the reasoning for introducing the BRP into the model is based on physical grounds, it is not clear whether its application is justified and leads to more realistic model results in the present case. However, the goal here is not necessarily to improve model results but to test how sensitive the model results react to a vertical redistribution of rejected salt. We here compare brine tracer concentrations, which correspond to the salinity of the rejected salt and are subject to the vertical distribution of the BRP, allowing to follow the pathways of water masses at the prescribed salt plume depth.

The effect of the BRP can be seen from Figure 8, showing the mean BoS brine tracer concentration in March along the zonal transect in the Bothnian Sea in the reference experiment and the two BRP experiments. While in the reference experiment, the tracer is vertically homogeneously distributed within the mixed layer, in the BRP bottom experiment, the tracer concentration is gradually increasing with depth and is maximal at the bottom. The differences between the reference experiment and the BRP MLD experiment are less obvious. In both cases, the tracer spreads only within the surface mixed layer, but in the BRP MLD experiment, stronger vertical gradients are found, and at the eastern coast, some tracer sinks below the mixed layer depth. The weakness of these differences is due to a rather conservative choice of MLD criterion, meaning





Figure 6. Snapshots of the tracer concentrations of the three passive surface tracers (left column: BoB tracer, middle column: BoS sea tracer, right column: GoF tracer) at the bottom of the Baltic Sea at the end of the indicated month in the reference experiment.

that the tracer is only mixed down to the upper boundary of the MLD and is subject to surface mixing in a similar manner as in the reference experiment.

To answer the question whether localized brine rejection in sea ice leads might be able to form deep water within the northern sub-basins which then enters the deeper parts of the central Baltic Sea, we analyze the brine tracer concentration at Landsort Deep. The solid lines in Figure 9a show the mean profiles of brine



Figure 7. Time series of 30-day running mean tracer concentration of BoS and GoF surface tracers at Landsort Deep (station BY31) in the reference experiment. The BoB tracer is not shown here as no significant amounts of tracer (concentrations higher than $0.01 m^{-3}$) reach Landsort Deep within the time scale of the simulation period. Note that a logarithmic scale is chosen for better visualization.

tracer concentration (sum of all three tracers) at station BY31 averaged over the period from June to September (i.e., the period of strongest tracer inflow) in the reference and the two BRP experiments. In all experiments, the highest tracer concentrations are found near the surface. While the concentration profile in the BRP MLD experiment is hardly changed compared to the reference experiment, the surface inflow of tracer concentration is reduced by 40% in the BRP bottom experiment. This is expected as, due to the downward distribution of brine tracer, the tracer concentration in the surface water transported into the Baltic proper is reduced. An interesting feature appearing in the BRP bottom experiment is the increase of tracer concentration in a depth range of 60-90 m, which is shown in the inset of Figure 9a. A closer look at the contributions from the individual tracers, shown in Figure 9b, reveals that the origin of this deeper intrusion of tracer-marked water is the Gulf of Finland. From the tracer concentrations alone, it is not possible to distinguish whether the inflow in this greater depth range is a unique feature of the BRP bottom experiment as a result of the formation of more dense bottom water or if it is also present in the other experiments and only becomes visible by the marking of bottom water with tracer. However, a comparison of the temporal behavior of the inflow and the prevailing wind forcing (see supporting information) suggests that the inflow is most likely caused by a wind-driven reversal of estuarine circulation in the Gulf of Finland, which is possible if strong southwesterly winds with a wind stress exceeding 0.02-0.04 N/m² persist (Elken et al., 2003). One could, however, claim that even if the formation of denser bottom water through the BRP is not causing the inflow, surface water from the Gulf of Finland is transported into intermediate depths of the central Baltic Sea, assuming that it is brought down to the bottom beforehand by strong localized brine rejection.

Despite the appearance of the additional deeper inflow of bottom water from the Gulf of Finland in the BRP bottom experiment, a contribution to the Landsort Deep bottom waters from the northern sub-basins, as hypothesized by Moros et al. (2020), is not found. For this to be possible, two requirements must be met: first,









Figure 9. Vertical profiles of mean brine tracer concentration (averaging period: June–September) at Landsort Deep (station BY31). (a) Sum of all three tracers in all model experiments. (b) Contribution of BoS and GoF tracer in BRP bottom experiments.

a water mass with a density high enough to be layered into the central Baltic Sea bottom waters must be formed in the sub-basins, and second, the circulation conditions must be such that this water mass is transported into the central Baltic Sea. Even though in the model, no inflow of bottom water from the Gulf of Bothnia into the central Baltic Sea is occurring, we can test whether the increase in bottom water salinity in the BRP bottom experiment would in principle be sufficient to form a water mass dense enough to contribute to Landsort Deep sub-halocline waters. Figure 10 shows the distribution of winter mean bottom salinity differences between the BRP bottom and the reference experiment in the grid cells of the Bothnian Bay, Bothnian Sea, and Gulf of Finland. On average, the winter bottom salinity is increased by 0.088 g/kg in the Bothnian Bay, 0.083 g/kg in the Bothnian Sea, and 0.054 g/kg in the Gulf of Finland due to the BRP. Locally and on shorter (daily instead of seasonal) time scales, the increase of bottom salinity can be much higher with up to 3 g/kg. However, this is only the case in shallow coastal areas, such as the Archipelago Sea. Assuming that a water mass which could potentially contribute to Landsort Deep bottom water ventilation would require a salinity of at least 9 g/kg, in both reference and BRP bottom experiment, only the Gulf of Finland bottom water is eligible. Considering the salinity bias of the model at Landsort Deep (see supporting information), the inflow of Gulf of Finland bottom water (Figure 9) could occur at greater depths in reality than it is the case in the model. Nevertheless, the process of localized brine rejection itself is too weak to form a dense water mass in the northern sub-basins which could potentially ventilate the northern central Baltic Sea.

3.4. Cold Climate Conditions

In order to answer the question whether the situation changes in a colder climate state, we tested the sensitivity of the results to the atmospheric forcing by repeating the previous experiments with air temperatures



Figure 10. Distribution of bottom salinity differences in winter (JFM mean) between the BRP bottom and the reference experiment in the grid cells of the Bothnian Bay, Bothnian Sea, and Gulf of Finland sub-basins.



reduced by 2 K. The temperature forcing modifies the surface buoyancy flux due to additional cooling of the surface water as well as increased ice growth and associated brine rejection.

The Cold Climate experiment is equivalent to the reference experiment except for the atmospheric forcing. From comparing the sinking of surface tracer between these two experiments, we can draw conclusions on possible changes in wintertime deep water formation in the northern sub-basins. The dashed lines in Figure 5 show vertical profiles of winter mean surface tracer concentration at selected stations in each of the sub-basins, similar to the solid lines for the reference experiment. In the Gulf of Finland and the Bothnian Sea as well as at station F9 in the Bothnian Bay, the profiles are almost unchanged with respect to the reference experiment. At station BO3, which was the only station with ground-reaching deep water formation in the reference experiment, the profile is modified, exhibiting lower tracer concentrations at sub-halocline depths except for the bottom layer where tracer concentrations are slightly increased. Thus, apart from some modulation due to the different buoyancy forcing, the emergence of deep water formation is qualitatively unchanged in the Cold Climate experiment.

Apart from the changes in local deep water formation, we are interested in changes in the inflow of tracer-marked water from the northern sub-basins into the Baltic proper. The dashed lines in Figure 9 show the mean brine tracer concentration at Landsort Deep in the Cold Climate experiments compared to the equivalent winter 1986/1987 experiments (solid lines). In all of the Cold Climate experiments, more brine tracer is arriving at Landsort Deep than in their respective reference experiments. This is because, due to the lower air temperatures, more sea ice is formed, and hence, more brine tracer is created. The depth ranges of brine tracer inflow are, however, unchanged in the Cold Climate experiments and the main inflow is still within the surface layer. In the BRP bottom experiment, where the brine tracer is brought to the bottom of the sea, the brine tracer inflow in the surface layer is increased only very little by the Cold Climate forcing. The deeper inflow observed in this experiment is still present and more than doubled in strength with the Cold Climate forcing. However, it is not found at greater depths, as one could expect due to a formation of denser water, but peaks even at slightly lower depths.

Summarizing, despite the increased sea ice formation and modified buoyancy forcing, the Cold Climate experiments yield qualitatively similar results to their corresponding winter 1986/1987 experiments.

4. Discussion

The model experiments conducted in this study suggest that the process of convective wintertime deep water formation plays no major role for bottom water ventilation in the northern and central Baltic Sea, which is in line with the established view (Marmefelt & Omstedt, 1993). By analyzing deep water formation in the individual northern Baltic Sea sub-basins through passive tracer experiments, the southern central part of the Bothnian Bay (in the vicinity to station BO3) was identified as a region where winter surface water from the same basin is brought down to the bottom of the sea. Although brine rejection along the eastern coast might have enhanced this deep water formation, it is primarily attributable to an inflow of denser Bothnian Sea surface water, mixing with Bothnian Bay surface water and sinking to the bottom following the basin topography. This finding is in agreement with the conclusions drawn by Neumann et al. (2020) based on ship-based measurements. In the Bothnian Sea and the deep parts of the Gulf of Finland, no bottom water renewal by winter surface water could be detected in the reference experiment. Hence, there is also no lateral inflow of deep water formed during wintertime in the northern basins ventilating the northern central Baltic Sea, as hypothesized to occur in cold climate periods by Moros et al. (2020). Instead, the inflow from the northern basins into the Baltic proper is restricted to the surface layer above the halocline at around 60 m. These results are robust to changing the atmospheric forcing to colder climate conditions by reducing the air temperature by 2 K. The introduction of a BRP, mixing all salt released during ice formation to the ground (BRP bottom experiment), is likely to overestimate the actual effect of localized brine rejection but leads to deep water formation in all northern sub-basins. As a result, deep water formed in the Gulf of Finland ventilates the Landsort Deep in a depth range of about 60-90 m. Due to a salinity bias at Landsort Deep (see supporting information), the inflow of Gulf of Finland bottom water might occur at even greater depths in reality. Although only appearing in the BRP bottom experiment, the inflow can most likely be attributed to a wind-driven estuarine circulation reversal (Elken et al., 2003) and not to a circulation change due to brine rejection. The effect of localized brine rejection has no major impact on the deep water exchange between the Gulf of Bothnia and the Baltic proper and is too weak to create a dense water mass which could potentially contribute to a ventilation of Landsort Deep bottom waters.

Similar results were found by performing comparable simulations with the ocean circulation model GETM (Burchard & Bolding, 2002) by Giesse (2018). The robustness of the results throughout various sensitivity experiments and models strongly suggests that convective wintertime deep water formation is not a relevant process for the ventilation of the northern and central Baltic Sea, at least in present climate conditions. Still, there is clear geological evidence for the occurrence of this process during cold climate periods (Moros et al., 2020). It could, therefore, be that the chosen approach, based on an ocean circulation model, is not suitable to fully and realistically simulate the involved processes or that further factors, not taken into account in our simulations, are crucial for deep water formation in the northern Baltic Sea. We discuss important limitations of our approach in the following.

4.1. Salinity Bias

As the convective sinking of water masses is mainly determined by the stratification conditions in the water column, a good representation of the hydrographic conditions in the model is crucial for this study. While the MOM ocean circulation model reproduces observed temperatures and surface salinities well, bottom water salinities are overestimated in the Bothnian Sea and the northern Baltic proper, in particular at Landsort Deep, where modeled bottom salinities exceed the observed salinities by approximately 2 g/kg (see supporting information). This, in turn, leads to unrealistically high stratification in the respective basins. As a consequence, the depth of convective sinking might be underestimated at these locations and also inflows into the Baltic proper might layer into too shallow depths due to the inaccurate density profile. However, we could show that, even with ground-reaching brine rejection, the salinity increase in the northern basins is not sufficient to reach salinities comparable to observed salinities in the great depths of Landsort Deep. We therefore do not expect the salinity biases to qualitatively change the general conclusions drawn from the model results. Furthermore, one should note that the usage of a different ocean circulation model would not solve the issue, as other models have similar difficulties in realistically representing salinity and other tracer concentrations in the northern Baltic Sea, as pointed out by Placke et al. (2018).

4.2. Bathymetry and Sill Depths

Circulation patterns and water exchange between different basins of the Baltic Sea are strongly impacted by the seafloor topography, especially by bathymetric sills (e.g., Laanearu & Lundberg, 2003; Lass & Mohrholz, 2003). Prominent bathymetric features can be localized on scales much smaller than the model grid size. As a consequence, those features are smoothed out and replaced by an average depth per grid cell in the model bathymetry.

The deep water exchange between the northern Baltic proper and the Bothnian Sea is of particular interest for our study. It occurs through the Åland Sea, which consists of two basins and three sills with depths of 100 (northern Åland sill) and 70 m (middle and southern Åland sills) (Leppäranta & Myrberg, 2009). Jakobsson et al. (2019) provide a detailed comparison of the Åland Sea bottom topography in high-resolution multibeam data from the R/V Electra and two digital bathymetric models (DBMs): IOWTOPO (resolution of 2×1 arcmin, Seifert et al., 2001) and the more recent EMODnet (resolution of $1/16 \times 1/16$ arcmin, EMODnet Bathymetry Consortium, 2018). While the main morphology of the Åland Sea is well represented in EMODnet compared to the high-resolution data from R/V Electra, the coarser resolved IOWTOPO DBM, on which our model bathymetry is based, only gives a rough picture of the highly complex Åland Sea bottom topography. Deviations in depth of up to 100 m are found in some locations. The sills are located nearly at the same locations as in EMODnet. However, they are substantially shallower. The northern sill has a depth of 57 m in IOWTOPO (compared to 88 m in EMODnet and 100 m reported by Leppäranta & Myrberg, 2009) and the middle and southern sill have a depth of 49 m in IOWTOPO (compared to 60 m in EMODnet and 70 m reported by Leppäranta & Myrberg, 2009). Furthermore, the middle sill is represented by only one deep passage in IOWTOPO instead of three passages identified in EMODnet. The depths of the sills separating the Bothnian Sea and the Bothnian Bay are likely to be equally underestimated in IOWTOPO. The Gulf of Finland, however, is not separated from the Baltic proper by any sills (Leppäranta & Myrberg, 2009).

It is of course not possible to represent the bottom topography on scales smaller than the model grid size. However, the loss of structure and the underestimation of sill depths could influence and possibly restrict



the deep water exchange in the model. It is conceivable that in reality, small-scale bottom currents that are highly dependent on topographic features can form and exchange deep water between the Gulf of Bothnia and the central Baltic Sea. Those bottom currents could be too small-scaled to be resolved in a typical regional ocean circulation model but would imprint to the seafloor morphology, for instance as erosion channels or drift deposits as found by Moros et al. (2020) and also in the high-resolution multibeam data from R/V *Electra* presented in Jakobsson et al. (2019).

4.3. Unresolved Processes

A crucial point in determining the potential of brine rejection in causing deep water formation is the degree of mixing of the rejected salt with the ambient water. Does the rejected highly saline brine sink down to the bottom or does it mix within the surface layer? When performing simulations with an ocean circulation model, one has to rely on the implemented vertical mixing scheme to realistically capture the mixing and convection. Here, the KPP scheme (Large et al., 1994) is used; in analogous GETM simulations (Giesse, 2018), a k- ϵ -turbulence closure model (Burchard et al., 1999) and vertically adaptive σ -coordinates are used (Burchard & Beckers, 2004; Hofmeister et al., 2010). While the mixing schemes can be assumed to work sufficiently well for homogeneous surface boundary conditions leading to widespread convection, difficulties arise when the surface boundary conditions are heterogeneous; that is, the buoyancy flux varies on scales smaller than the horizontal resolution of the model (e.g., Losch et al., 2006). The averaging of the buoyancy flux over the grid cell reduces the actual potential of deep-reaching, local convection. Here, the localized convection has been accounted for by performing sensitivity experiments with a subgrid-scale BRP adopted from Nguyen et al. (2009). This parameterization has some shortcomings. For example, it neglects the mixing of heat, momentum, and tracers other than salt and should not be applied under certain conditions (e.g., for widespread convection or high ice-ocean relative velocities) (Barthélemy et al., 2015). It is therefore not clear whether the introduction of a BRP leads to improved model results in this particular case. Irrespective of this, the sensitivity experiments with BRP show that the degree of mixing of rejected brine with ambient water does not impact the large-scale circulation and deep water exchange between the northern sub-basins and the Baltic proper in the model.

Nevertheless, the fact that current, state-of-the-art ocean circulation models tend to have difficulties in realistically representing the hydrographic conditions in the northern Baltic Sea (Placke et al., 2018) supports the hypothesis of processes being not or not well represented in the models for this particular region. The issue of a low-resolution bathymetry, as discussed above, could also add to or be the primary reason for the model biases. As the modeling of convection and deep water formation has proven difficult also in other regions in the world (e.g., the northwestern Mediterranean Sea; Somot et al., 2016), a better understanding of the processes taking place in the northern Baltic Sea might be needed. High-resolution and non-hydrostatic model simulations might give more conclusive insights.

4.4. Restricted Simulation Period

The analysis and the sensitivity experiments are restricted to the period of one particular winter and the following months. We chose the year 1986/1987 because it has been one of the coldest winters in the more recent climate period, with the largest ice extent since the simulation start in 1950. We would therefore expect the largest impact of brine rejection in this winter. However, when it comes to circulation changes, other factors such as the wind forcing, which can vary substantially from year to year, also have a strong impact. Moreover, the sensitivity experiments lack a spin-up period and it might take much longer for the circulation to fully adapt to the different atmospheric forcing or the application of a BRP. Longer simulations might be more conclusive, but would also be computationally more expensive.

4.5. Paleosalinity

Our model simulations are based on the present climate state. In the Cold Climate experiments, we have modified the air temperature forcing only. However, also other forcing factors which impact the hydrographic conditions could have been different during colder climate periods like the LIA. Paleoenvironmental reconstructions indicate that the salinity of the Baltic Sea was lower during the LIA compared to the present climate period (Emeis et al., 2003; Leipe et al., 2008), which might have gone along with a less pronounced halocline separating the surface and bottom water masses. Giesse (2018) has shown that in the extreme scenario of an unstratified Baltic Sea, with homogeneous initial conditions for



temperature and salinity, brine rejection would indeed lead to deep water formation in the northern sub-basins and cause an inflow of these dense bottom waters into the Baltic proper. Although the scenario of an unstratified Baltic Sea is only hypothetical, it is very likely that the actual hydrographic conditions, in particular during the LIA period, were less stratified than represented in the model, which could have facilitated deep water formation.

5. Summary and Conclusions

In this study, we revisited the role of convective deep water formation in Baltic Sea deep water renewal by analyzing passive tracer pathways in several sensitivity experiments based on the severe winter 1986/1987 with the ocean circulation model MOM. In particular, we assessed the potential effect of localized brine rejection by applying a BRP (Nguyen et al., 2009) to the Baltic Sea for the first time. Generally, our model results support the established view that convective deep water formation does not play a major role in Baltic Sea deep water renewal. We find no evidence for deep water formation caused by thermal or haline convection in the northern sub-basins and, hence, also no ventilation of the northern Baltic proper through deep water formed in the sub-basins during wintertime. These results are robust to changing the atmospheric forcing to colder climate conditions by reducing the air temperature. Localized brine rejection could potentially lead to deep water formation. However, the effect is too weak to impact the large-scale circulation and deep water exchange between the Baltic proper and the sub-basins and to create a substantial dense water mass in the Bothnian Sea, which could possibly ventilate the northern Baltic proper. Solely, bottom water from the Gulf of Finland is found to contribute to the ventilation of the Landsort Deep at sub-halocline depths, caused, however, by a wind-driven reversal of the estuarine circulation.

The used model approach has some shortcomings, such as a salinity bias leading to overestimated stratification in the northern Baltic Sea, non-resolved bathymetric features influencing the deep water exchange, and a general limitation in representing small-scale processes due to resolution. Moreover, we have restricted the simulation period to one specific winter as a case study and have not taken into account potentially different hydrographic conditions during cold climate periods like the LIA. Nevertheless, we would like to stress that models with different vertical coordinates and different mixing schemes lead to the same conclusion that under present climate, deep water formation would be too weak to counteract the estuarine circulation in the northern Baltic Sea (Giesse, 2018).

Still, geological findings suggest wintertime deep water formation to have contributed to deep water renewal in the northern Baltic Sea during cold climate phases (Moros et al., 2020) and there is clear evidence for a better ventilation during the LIA (e.g., Kabel et al., 2012; Leipe et al., 2008). Considering the model limitations, we can gain insights on how and under which circumstances convective deep water formation could potentially have occurred and ventilated the Baltic Sea bottom waters. It is conceivable that in reality, deep water formation is associated with the formation of small-scale bottom currents, which cannot be resolved in the model. These bottom currents could leave the Gulf of Bothnia through the western Åland Sea following the bathymetry. Larger-scale ventilation during the LIA could be explained by not only reduced temperatures but also reduced salinity going along with weaker stratification conditions, which would be in agreement with paleoenvironmental studies such as Emeis et al. (2003) and Leipe et al. (2008). A different wind forcing and increased storminess during the LIA could further impact the ventilation (Meier, 2005; Zillén & Conley, 2010).

Bringing together the geological evidence for ventilation of the northern Baltic Sea through wintertime deep water formation and ocean model simulations would be an important contribution to improve process understanding in the northern Baltic Sea in different climate periods. Since wintertime observations in the northern Baltic Sea are scarce and information about past climate periods that can be retrieved from geological and paleontological studies is limited, the role of models for process understanding is crucial. The fact that various state-of-the-art ocean models have difficulties to realistically represent the hydrographic conditions in the northern Baltic Sea (Placke et al., 2018) emphasizes the need for improved process understanding and representation in models. The question of ventilation processes in the Baltic Sea is particularly interesting in the context of hypoxia in different climate periods (e.g., Jilbert et al., 2015; Papadomanolaki et al., 2018; Zillén et al., 2008). Having shown that, in our model approach, accounting for localized brine rejection and reduced air temperature is not sufficient to cause a ventilation of the northern Baltic Sea



through wintertime deep water formation, further research is needed. Sensitivity experiments with different salinity and stratification conditions for the LIA and/or high-resolution process studies analyzing, for example, the deep water exchange through the complex bathymetry of the Åland Sea could bring more insights.

Data Availability Statement

The model data generated and analyzed in this study are available in a repository (https://thredds-iow.io-warnemuende.de/thredds/catalogs/publications/thomas/catalog_giesse2020.html).

References

- Andrejev, O., Myrberg, K., Alenius, P., & Lundberg, P. A. (2004). Mean circulation and water exchange in the Gulf of Finland: A study based on three-dimensional modelling. *Boreal Environment Research*, 9(1), 1–16.
- Barthélemy, A., Fichefet, T., Goosse, H., & Madec, G. (2015). Modeling the interplay between sea ice formation and the oceanic mixed layer: Limitations of simple brine rejection parameterizations. *Ocean Modelling*, *86*, 141–152. https://doi.org/10.1016/j.ocemod.2014.12.009 Barthélemy, A., Fichefet, T., Goosse, H., & Madec, G. (2016). A multi-column vertical mixing scheme to parameterize the heterogeneity of
- oceanic conditions under sea ice. *Ocean Modelling*, 104, 28–44. https://doi.org/10.1016/j.ocemod.2016.05.005
- Burchard, H., & Beckers, J.-M. (2004). Non-uniform adaptive vertical grids in one-dimensional numerical ocean models. *Ocean Modelling*, 6(1), 51–81. https://doi.org/10.1016/S1463-5003(02)00060-4

- Burchard, H., Bolding, K., & Ruiz-Villarreal, M. (1999). GOTM, a general ocean turbulence model. Theory, implementation and test cases, Tech. Rep. EUR 18745 EN: European Commission.
- Duffy, P. B., & Caldeira, K. (1997). Sensitivity of simulated salinity in a three-dimensional ocean model to upper ocean transport of salt from sea-ice formation. *Geophysical Research Letters*, 24(11), 1323–1326. https://doi.org/10.1029/97GL01294

Duffy, P. B., Eby, M., & Weaver, A. J. (1999). Effects of sinking of salt rejected during formation of sea ice on results of an ocean-atmospheresea ice climate model. *Geophysical Research Letters*, 26(12), 1739–1742. https://doi.org/10.1029/1999GL900286

EMODnet Bathymetry Consortium (2018). EMODnet Digital Bathymetry (DTM 2018). https://doi.org/10.12770/18FF0D48-B203-4A65-94A9-5FD8B0EC35F6

Elken, J. A., Raudsepp, U., & Lips, U. (2003). On the estuarine transport reversal in deep layers of the Gulf of Finland. *Journal of Sea Research*, 49(4), 267–274. https://doi.org/10.1016/S1385-1101(03)00018-2

Emeis, K.-C., Struck, U., Blanz, T., Kohly, A., & Voß, M. (2003). Salinity changes in the central Baltic Sea (NW Europe) over the last 10000 years. The Holocene, 13(3), 411-421. https://doi.org/10.1191/0959683603hl634rp

Fischer, H., & Matthäus, W. (1996). The importance of the Drogden Sill in the Sound for major Baltic inflows. *Journal of Marine Systems*, 9(3), 137–157. https://doi.org/10.1016/S0924-7963(96)00046-2

- Geyer, B. (2014). High-resolution atmospheric reconstruction for Europe 1948–2012: coastDat2. Earth System Science Data, 6(1), 147–164. https://doi.org/10.5194/essd-6-147-2014
- Giesse, C. (2018). Numerical sensitivity study on haline convection due to sea ice brine rejection in the northern Baltic Sea (Master's Thesis). University of Rostock.
- Granqvist, G. (1938). Zur Kenntnis der Temperatur und des Salzgehaltes des Baltischen Meeres an den Küsten Finnlands. Societas Geographica Fenniae, 122.
- Griffies, S. M. (2004). Fundamentals of ocean climate models. Princeton, NJ: Princeton University Press.

Hofmeister, R., Burchard, H., & Beckers, J.-M. (2010). Non-uniform adaptive vertical grids for 3D numerical ocean models. Ocean Modelling, 33(1), 70–86. https://doi.org/10.1016/j.ocemod.2009.12.003

- Jakobsson, M., Stranne, C., O'Regan, M., Greenwood, S. L., Gustafsson, B., Humborg, C., & Weidner, E. (2019). Bathymetric properties of the Baltic Sea. Ocean Science, 15(4), 905–924. https://doi.org/10.5194/os-15-905-2019
- Jilbert, T., Conley, D. J., Gustafsson, B. G., Funkey, C. P., & Slomp, C. P. (2015). Glacio-isostatic control on hypoxia in a high-latitude shelf basin. Geology, 43(5), 427–430. https://doi.org/10.1130/G36454.1
- Jin, M., Hutchings, J., & Kawaguchi, Y. (2015). Sensitivity study of subgrid scale ocean mixing under sea ice using a two-column ocean grid in climate model CESM. Frontiers of Earth Science, 9(4), 594–604. https://doi.org/10.1007/s11707-014-0489-9
- Jin, M., Hutchings, J., Kawaguchi, Y., & Kikuchi, T. (2012). Ocean mixing with lead-dependent subgrid scale brine rejection parameterization in a climate model. *Journal of Ocean University of China*, 11(4), 473–480. https://doi.org/10.1007/s11802-012-2094-4

Kabel, K., Moros, M., Porsche, C., Neumann, T., Adolphi, F., Andersen, T. A. B. J., et al. (2012). Impact of climate change on the Baltic Sea ecosystem over the past 1,000 years. *Nature Climate Change*, 2(12), 871–874. https://doi.org/10.1038/nclimate1595

Kozo, T. L. (1983). Initial model results for Arctic mixed layer circulation under a refreezing lead. *Journal of Geophysical Research*, 88(C5), 2926–2934. https://doi.org/10.1029/JC088iC05p02926

Laanearu, J., & Lundberg, P. (2003). Topographically constrained deep-water flows in the Baltic Sea. *Journal of Sea Research*, 49(4), 257–265. https://doi.org/10.1016/S1385-1101(03)00037-6

Large, W. G., McWilliams, J. C., & Doney, S. C. (1994). Oceanic vertical mixing: A review and a model with a nonlocal boundary layer parameterization. *Reviews of Geophysics*, 32(4), 363–403. https://doi.org/10.1029/94RG01872

Lass, H. U., & Mohrholz, V. (2003). On dynamics and mixing of inflowing saltwater in the Arkona Sea. Journal of Geophysical Research, 108(C2), 3042. https://doi.org/10.1029/2002JC001465

Leipe, T., Dippner, J. W., Hille, S., Voss, M., Christiansen, C., & Bartholdy, J. (2008). Environmental changes in the central Baltic Sea during the past 1000 years: Inferences from sedimentary records, hydrography and climate. *Oceanologia*, 50(1), 23–41.

Leppäranta, M., & Myrberg, K. (2009). Physical oceanography of the Baltic Sea: Springer Science & Business Media.

Losch, M., Herlufsen, S., & Timmermann, R. (2006). Effects of heterogeneous surface boundary conditions on parameterized oceanic deep convection. Ocean Modelling, 13(2), 156–165. https://doi.org/10.1016/j.ocemod.2005.12.003

Marmefelt, E., & Omstedt, A. (1993). Deep water properties in the Gulf of Bothnia. Continental Shelf Research, 13(2–3), 169–187. https://doi.org/10.1016/0278-4343(93)90104-6

Acknowledgments

The research presented in this study is part of the Baltic Earth program (Earth System Science for the Baltic Sea Region, see https://www.baltic.earth/). The model development and simulation of MOM were performed with resources provided by the North-German Supercomputing Alliance (HLRN). Updates of the ice thickness time series used for validation purposes were provided by the ice service of the Finnish Meteorological Institute and post-processing of BED data for the validation of vertical profiles was performed by Manja Placke. Open access funding enabled and organized by Projekt DEAL.

Burchard, H., & Bolding, K. (2002). GETM, A general estuarine transport model: Scientific documentation, Tech. Rep. EEUR 20253 EN: European Commission.



- Matthäus, W., & Franck, H. (1992). Characteristics of major Baltic inflows a statistical analysis. Continental Shelf Research, 12(12), 1375–1400. https://doi.org/10.1016/0278-4343(92)90060-W
- Maykut, G. A. (1978). Energy exchange over young sea ice in the central Arctic. Journal of Geophysical Research, 83(C7), 3646–3658. https:// doi.org/10.1029/JC083iC07p03646

Meier, H. E. M. (2005). Modeling the age of Baltic seawater masses: Quantification and steady state sensitivity experiments. *Journal of Geophysical Research*, 110, C02006. https://doi.org/10.1029/2004JC002607

- Meier, H. E. M., Feistel, R., Piechura, J., Arneborg, L., Burchard, H., Fiekas, V., et al. (2006). Ventilation of the Baltic Sea deep water: A brief review of present knowledge from observations and models. *Oceanologia*, 48(S).
- Morison, J. H., & McPhee, M. G. (1998). Lead convection measured with an autonomous underwater vehicle. Journal of Geophysical Research, 103(C2), 3257–3281. https://doi.org/10.1029/97JC02264
- Morison, J. H., McPhee, M. G., Curtin, T. B., & Paulson, C. A. (1992). The oceanography of winter leads. Journal of Geophysical Research, 97(C7), 11,199–11,218. https://doi.org/10.1029/92JC00684
- Moros, M., Kotilainen, A. T., Snowball, I., Neumann, T., Perner, K., Meier, H. E. M., et al. (2020). Is deep-water formation in the Baltic Sea a key to understanding seabed dynamics and ventilation changes over the past 7,000 years? *Quaternary International*, 550, 55–65. https://doi.org/10.1016/j.quaint.2020.03.031
- Muench, R. D., Smith, D. C., & Paulson, C. A. (1995). Convection beneath freezing leads: New observations compared with numerical model results. *Journal of Geophysical Research*, 100(C3), 4681–4692. https://doi.org/10.1029/94JC02503

Myrberg, K., & Andrejev, O. (2006). Modelling of the circulation, water exchange and water age properties of the Gulf of Bothnia. Oceanologia, 48(S), 55–74.

Neumann, T., Radtke, H., & Seifert, T. (2017). On the importance of major Baltic inflows for oxygenation of the central Baltic Sea. Journal of Geophysical Research, 122, 1090–1101. https://doi.org/10.1002/2016JC012525

- Neumann, T., Siegel, H., Moros, M., Gerth, M., Kniebusch, M., & Heydebreck, D. (2020). Ventilation of the northern Baltic Sea. Ocean Science, 16(4), 767–780. https://doi.org/10.5194/os-16-767-2020
- Nguyen, A. T., Menemenlis, D., & Kwok, R. (2009). Improved modeling of the Arctic halocline with a subgrid-scale brine rejection parameterization. Journal of Geophysical Research, 114, C11014. https://doi.org/10.1029/2008JC005121
- Notz, D., & Worster, M. G. (2009). Desalination processes of sea ice revisited. Journal of Geophysical Research, 114, C05006. https://doi.org/ 10.1029/2008JC004885
- Papadomanolaki, N. M., Dijkstra, N., van Helmond, N. A. G. M., Hagens, M., Bauersachs, T., Kotthoff, U., et al. (2018). Controls on the onset and termination of past hypoxia in the Baltic Sea. *Palaeogeography Palaeoclimatology Palaeoecology*, 490, 347–354. https://doi.org/ 10.1016/j.palaeo.2017.11.012

Placke, M., Meier, H. E. M., Gräwe, U., Neumann, T., Frauen, C., & Liu, Y. (2018). Long-term mean circulation of the Baltic Sea as represented by various ocean circulation models. *Frontiers in Marine Science*, *5*, 287. https://doi.org/10.3389/fmars.2018.00287

Schinke, H., & Matthäus, W. (1998). On the causes of major Baltic inflows an analysis of long time series. Continental Shelf Research, 18(1), 67–97. https://doi.org/10.1016/S0278-4343(97)00071-X

Seifert, T., Tauber, F., & Kayser, B. (2001). A high resolution spherical grid topography of the Baltic Sea - 2nd edition. Baltic Sea Science Congress, Stockholm, 25–29 November 2001, Poster #147.

- Smagorinsky, J. (1963). General circulation experiments with the primitive equations. *Monthly Weather Review*, 91(3), 99–164. https://doi. org/10.1175/1520-0493(1963)091<0099:GCEWTP>2.3.CO;2
- Smith, D. C., Lavelle, J. W., & Fernando, H. J. S. (2002). Arctic Ocean mixed-layer eddy generation under leads in sea ice. Journal of Geophysical Research, 107(C8), 17–1. https://doi.org/10.1029/2001JC000822
- Smith, D. C., & Morison, J. H. (1993). A numerical study of haline convection beneath leads in sea ice. Journal of Geophysical Research, 98(C6), 10,069–10,083. https://doi.org/10.1029/93JC00137
- Smith, D. C., & Morison, J. H. (1998). Nonhydrostatic haline convection under leads in sea ice. Journal of Geophysical Research, 103(C2), 3233–3247. https://doi.org/10.1029/97JC02262
- Somot, S., Houpert, L., Sevault, F., Testor, P., Bosse, A., Taupier-Letage, I., et al. (2016). Characterizing, modelling and understanding the climate variability of the deep water formation in the north-western Mediterranean Sea. *Climate Dynamics*, 51(3), 1179–1210. https:// doi.org/10.1007/s00382-016-3295-0

Untersteiner, N. (1968). Natural desalination and equilibrium salinity profile of perennial sea ice. Journal of Geophysical Research, 73(4), 1251–1257. https://doi.org/10.1029/JB073i004p01251

- Winton, M. (2000). A reformulated three-layer sea ice model. Journal of Atmospheric and Oceanic Technology, 17(4), 525–531. https://doi. org/10.1175/1520-0426(2000)017<0525:ARTLSI>2.0.CO;2
- Zillén, L., & Conley, D. J. (2010). Hypoxia and cyanobacteria blooms—Are they really natural features of the late Holocene history of the Baltic Sea?. *Biogeosciences*, 7(8), 2567–2580. https://doi.org/10.5194/bg-7-2567-2010
- Zillén, L., Conley, D. J., Andrén, T., Andrén, E., & Björck, S. (2008). Past occurrences of hypoxia in the Baltic Sea and the role of climate variability, environmental change and human impact. Earth-Science Reviews, 91(1), 77–92. https://doi.org/10.1016/j.earscirev.2008.10.001